Author’s response to reviews

Title: The Utility of Dual Energy Computed Tomography in the Management of Axial Gout: Case Reports and Literature Review

Authors:

Jeremy Wang (jeremy.wang.syd@gmail.com)

Beverly Ng (Beverly.ck.ng@gmail.com)

Haesung Bak (haesungbak@bigpond.com)

David Spencer (davidsp@bigpond.com)

Nicholas Manolios (nicholas.manolios@sydney.edu.au)

Peter Wong (peter.wong2@health.nsw.gov.au)

Version: 2 Date: 06 Feb 2020

Author’s response to reviews:

Dear Editor

Thank you for your time and interest in our manuscript. I have made changes according to your comments. Please find the details to each comments below

Editor Comments:

1. Ethics approval and consent to participate

Please include a statement on ethics approval and consent to participate in the “Ethics approval and consent to participate” section of the Declarations. This section must be included even if it is not applicable to your manuscript. If it is not applicable to your manuscript please write ‘Not Applicable’ in this section.

- This has been added to the final manuscript with the rest of declarations and ordered in the desired way as recommended.

2. Headings

Please insert the heading 'Declarations' for the Declarations section, below the 'Abbreviations' section of your manuscript.
- This has been added to the final manuscript.

3. Additional files

Thank you for providing a completed CARE checklist. As this document is no longer required at this stage of the publication process, please remove this from your submission’s supplementary files.

- This has been deleted.

4. Clean copy

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

- The final copy has been uploaded with two figures uploaded as separate files.

Thank you again for your time. We look forward to your response.

Sincerely

Jeremy Wang

Rheumatology Advanced Trainee